Author's response to reviews

Title: Teams and Quality of Care in Nursing Homes

Authors:

Anders Kvale Havig (akh@nova.no)  
Anders Skogstad (anders.skogstad@psysp.uib.no)  
Marijke Veenstra (marijke.veenstra@nova.no)  
Tor Inge Romøren (tromoren@online.no)

Version: 2 Date: 30 April 2013

Author's response to reviews:

Cover letter

*Requesting consent statement – This is included, se page 10. Since none of the questions contained were about the residents and it was voluntary to participate in the study, we judged that a consent statement was not necessary. Furthermore, prior to the study we asked the The Norwegian Directorate of Health and the Norwegian Board of Health Supervision if this was acceptable, and both institutions accepted this approach.

*Competing interests – This is included, see page 21

*Authors’ contributions – This is corrected, se page 21.

*Acknowledgements – This is corrected, se page 21.